Author's response to reviews

Title: Hutchinson-Gilford progeria syndrome accompanied by severer skeletal abnormalities in two Chinese siblings: a case series

Authors:

Zhimin Xiong (xiongzhimin@sklmg.edu.cn)
Yanmei Lu (luyanmei@sklmg.edu.cn)
Jinjie Xue (xuejinjie@sklmg.edu.cn)
Sanchuan Luo (luosanchuan@sklmg.edu.cn)
Xiaojian Xu (xuxiaojian@sklmg.edu.cn)
Lusi Zhang Zhang (zhuanglusi@sklmg.edu.cn)
Hao Peng (penghao@sklmg.edu.cn)
Wei Li (liwei@sklmg.edu.cn)
Dengming Chen (xyyyfuk@163.com)
Zhengmao Hu (hu Zhengmao@sklmg.edu.cn)
Kun Xia Dr. (xiakun@sklmg.edu.cn)

Version: 3 Date: 11 January 2013

Author's response to reviews: see over
Dear editor,

Thanks for your consideration on the revision of our manuscript titled “Hutchinson-Gilford progeria syndrome accompanied by severer skeletal abnormalities in two Chinese siblings: a case series” (MS: 2054895973787009). Now, we are submitting our revised manuscript. In the revised version, we have divided the case presentation in the abstract section, and the term female and male in describing the patients were used accordingly.

I am looking forward to hearing from you and reviewers.

Sincerely yours,

Kun Xia